A Systematic Review of Mobile-Assisted Language Learning Research Trends and Practices in Malaysia

https://doi.org/10.3991/ijim.v16i05.28129

Nursufyana Ulfa Che Mustaffa^(\boxtimes), Siti Nazuar Sailin Universiti Utara Malaysia, Sintok, Malaysia n.sufyanaulfa@gmail.com

Abstract-Mobile technology integration in Malaysia's English language education has a bright future, owing to the many advantages offered by mobile devices. However, there are inadequate studies that systematically review existing work in Malaysia to uncover gaps in the use of mobile technologies for English teaching and learning. Thus, a systematic review was conducted to offer an up-to-date synthesis of Mobile-Assisted Language Learning (MALL) integration in Malaysia from January 2019 to January 2021. The review followed the PRISMA Statement's guidelines which resulted in the identification of eleven articles that met all the inclusion criteria. Following that, the eleven articles were analysed and synthesised quantitatively and qualitatively using seven codes derived from two earlier systematic reviews of the literature. The codes were: (1) research objectives, (2) research methodologies, (3) educational level, (4) learning context, (5) device types, (6) theories, and (7) research outcomes. The systematic review discovered several significant findings, including the fact that the majority of the studies examined the effect of MALL integration on students' English acquisition and performance, and that MALL was more prevalent in Malaysian higher education institutions than other levels of education, with the lowest prevalence in primary schools. In light of this information, recommendations for future research were proposed. Malaysian academics are encouraged to broaden the scope of their future research and engage students from all levels of education in future MALL practices in order to better understand MALL's usability in English language instruction in Malaysia.

Keywords—systematic literature review, research trends, mobile-assisted language learning, MALL, english as second language (ESL), mobile technologies, Malaysia

1 Introduction

The advancement of the latest technologies has paved the way for multi-functional mobile devices such as smartphones, tablet PCs, pocket computers and e-readers, that make working and communicating facile and learning a second language simpler,

quicker, easier, and more enjoyable. The integration of the newest technologies into the framework of teaching and learning a language has benefitted both teachers and students because it has enabled them to develop and experience previously unimaginable methods of teaching and learning a second language.

Even though the spread of new technology, especially mobile phones, has evolved rapidly in both developed and developing countries, the differences in practices are staggering. Like all other technical advances, the affordability and availability of mobile learning are generally more successfully integrated into practices in developed countries, which have a different state of technology and social system than developing countries [1], [2]. Looking at the integration of mobile technology in teaching and learning a second language, it is evident that Malaysia is trailing a decade behind the advancements made in developed countries such as Australia, America, and China. Evidence indicates that Malaysia only started adopting mobile-assisted language learning (MALL) in 2003 [3], even though developed countries had begun exploiting mobile technology in language learning since 1994 [4].

Although Malaysia started relatively later, the ubiquity of mobile technology in today's society and the proliferation of mobile devices in the Malaysian community have significantly impacted the implementation of mobile technology in the education system, particularly in English language learning. [5] from Statista, who surveyed in 2019, noted that the number of smartphone users in Malaysia will reach more than 33 million by 2021. It was also recorded that the millennial generation is more likely to own smartphones than the older generation and that around 86.3% of respondents below the age of 20 own a smartphone [5]. The widespread ownership of smartphones among the millennial generation in Malaysia has influenced the pervasive integration of MALL in English language classes in Malaysia. However, as indicated by [6], there are still relatively few systematic literature reviews on this popular subject. In addition, there have been no comprehensive reviews on studies that have been published from January 2019 until January 2021, as the systematic literature review conducted by [6] only reviewed studies that were published from 2009 until 2019.

Consequently, the present systematic review is prompted by one fundamental research question: What research has been undertaken on the integration of MALL in language classrooms in Malaysia between January 2019 and January 2021? The main objective of the review is to provide an up-to-date summary of research trends and MALL practices in Malaysia between January 2019 to January 2021, based on seven codes obtained from previous systematic literature reviews: (1) research objectives, (2) research methodologies, (3) educational level, (4) learning context, (5) device types, (6) theories, and (7) research outcomes. The present paper is subdivided into six sections. The first section begins with an introduction. Section 2 provides a review of the literature on MALL in Malaysia and throughout the world. Section 3 explains the research methodology employed in this study. Following that, Section 4 discusses and integrates the findings and discussion with relevant literature. Section 5 highlights the gaps and gives recommendations for future studies. Finally, Section 6 summarises the entire review.

2 Literature review

Mobile device usage, particularly interactive touchscreen devices, has become ingrained and indispensable in the lives of today's younger generation, many of whom may have had their first experience with touchscreen devices prior to the age of two [7]. In an ideal situation, mobile devices, particularly touchscreen devices, would deliver interactive experiences that capitalise on a human's inherent constructivist learning capacities, thereby promoting the development of 21st-century skills deemed necessary for learning and future employment in an age of technology [8], [9]. Thus, according to [6], it is inevitable that the use of mobile technology, particularly touchscreen devices, as an effective tool for learning has become a global phenomenon, attracting significant attention from researchers, practitioners, and stakeholders across a range of educational contexts in an effort to advance towards Education 4.0.

The idea of mobile learning is not novel; thus, the definition of mobile learning has always been extended time and again. There are two major perspectives of mobile learning which explain the term from a technology-centred perspective and learner-centred perspective [10]. The earlier well-known definition of mobile learning originated from the technology-centred perspective proposed by [11, p. 262], who defined mobile learning as "any educational provision where the sole or dominant technologies are handheld or palmtop devices". [11, p. 263] characterised mobility of mobile learning as the use of mobile devices such as "phones, smartphones, personal digital assistants (PDAs) and their peripherals, perhaps tablet PCs and perhaps laptop PCs, but not desktops in carts and other similar solutions".

In addition, [12, p. 159] offered a different definition derived from the learner-centred perspective, by defining mobile learning as the portability of devices and the mobility of learners who "create ever-changing learning environments" that transcend time and places. In recent years, [13] presented a broader definition that best encapsulates the essence of mobile learning, by stating that mobile learning is the process of acquiring knowledge via intra- and inter-personal interactions among users as well as their learning materials in a range of scenarios through the use of personal mobile devices capable of swiftly collecting and delivering information to users.

2.1 Mobile-assisted language learning (MALL)

The constant evolution of mobile technology and associated technologies has resulted in the increased usage of mobile devices for language learning. MALL is a prominent subset of mobile learning that allows language educators to exploit the various features of mobile devices to assist in language acquisition [14]. The results from two research reviews, which were published five years apart, indicated that the typical sub-disciplines for the implementation of mobile learning in the field of education are still in the language and linguistic disciplines over 15 years, from 2000 to 2015 [15], [16]. The findings of these two research reviews suggested that mobile technology has been pervasively used by language educators and students to facilitate effective teaching and learning of the target language in the past and present and is most likely to have the same effect in the future. [17] and [18] noted that technological advances often impact education systems, wherein the future of learning appears to be increasingly mobile due to the evolution and functionalities of mobile technology. The integration of mobile technology into education has expanded the opportunities for students at all levels of the educational system to access innovative English language education [17], [18].

Furthermore, teaching and learning the English language through the use of mobile devices is becoming a common practice in many countries across various educational settings. MALL has been widely adopted in recent years in a number of countries, most particularly Indonesia [19], [20], Iran [21], [22], Japan [23], [24], Korea [25], [26], Malaysia [27], [28], Saudi Arabia [29], [30], Taiwan [31], [32], Thailand [33], [34], Turkey [35], [36], United States of America [37], [38], and Vietnam [39], [40]. The growing adoption trends of MALL indicate that students in both developed and developing countries utilise a range of mobile technologies for learning due to mobile technologies becoming more accessible and inexpensive globally.

Additionally, previous studies investigating students' and teachers' perceptions of the implementation of mobile technology in English language learning revealed that both groups have favourable perceptions and attitudes towards the use of MALL in teaching and learning English [29], [41], [42], [43], [44], [45]. Several studies concurred that the incorporation of MALL promotes constructive English language learning experiences in several areas of language skills relative to the conventional teaching and learning practices, such as vocabulary learning [46], reading skills [30], [47], writing skills [28], [43], speaking skills [48], [49], and listening skills [50], [51]. In addition, several studies also reported the positive effects of mobile technology on students' autonomy [24], [52] and motivation in learning English [53], [54]. MALL has rapidly grown over the past decade, and numerous studies conducted globally have proved that MALL has become ubiquitous in English language teaching and learning.

Nevertheless, amid the growing number of studies emphasising the positive effect on improving students' English language skills and profound acceptance and attitudes towards the use of MALL in English language learning, several studies also exposed the unfavourable aspects of the use of MALL in classrooms. These studies revealed several shortcomings, such as lack of students readiness and willingness to assume charge of their learning and manage flexibility in their schedules [22], difficulties in achieving learning targets due to the unavailability of knowledgeable and skilled individuals who can provide individualised attention and feedback [55], the limiting condition in learning owing to the limited features of mobile devices like small screen size and keys [56], and the detrimental effects of smartphones that cause learning distractions among students who use mobile technology in their learning [57], [58].

The emergence of the inconclusive findings demonstrates that the usefulness of MALL integration in the global language classroom requires further investigations and synthesis. Therefore, recent studies in this field must be reviewed systematically. This is to develop a comprehensive understanding and knowledge of recent findings, which will serve as valuable references for academics and practitioners, particularly in the Malaysian context. Furthermore, since new technology is continuously developing and

evolving, varying dramatically from two decades ago, researchers and practitioners must pay keen attention to the latest development in mobile technology as mobile devices can be prospective educational tools that can deliver and support learning content.

3 Methodology

The research method chosen for the investigation of the research question was the systematic literature review. A systematic literature review (SLR) is a systematic procedure that follows a specified process that allows researchers to explore the breadth and depth of existing literature that leads to the testing of new hypotheses, development of new theories, or identification of potential gaps that need to be explored [59]. The study's research method was chosen because it contributes to the consistent and concise synthesis of existing academic literature by confirming or refuting the literature's reliability, establishing the literature's quality, and addressing any uncertainties or variations in practice that may arise [60], [61]. This SLR was conducted to identify, summarise and evaluate the existing relevant articles on MALL implementation in the context of the education system in Malaysia that were published between January 2019 and January 2021.

3.1 Search strategy using PRISMA

The present SLR adhered to the guidelines provided by the PRISMA 2009 Statement. The PRISMA 2009 Statement is established to support systematic reviewers in reporting publicly why the review was undertaken, what the authors performed, and what they discovered [62]. The authors followed four major phases in PRISMA 2009 Statement, namely identification, screening, eligibility, and inclusion [63], to address a specific research question on MALL implementation in the Malaysian education system (see Figure 1).

3.2 Databases

Journal articles for this review were compiled from three extensively referenced and citation-enhanced electronic database indexing sites, including Elsevier's Scopus, Taylor & Francis, and ERIC. The Elsevier's Scopus and Taylor & Francis databases were selected because the scholarly community widely views the databases as authoritative sources of bibliometric data containing a large number of journal articles, research reports, conference papers, dissertations, and books published by respected journal publishers in a rigorous review process [64]. On the other hand, the ERIC database was selected because it is reliable and the most accessible educational database globally that is commonly used to search scholarly literature [65].



From: Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009). Preferred Reporting /tems for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6(7): e1000097. doi:10.1371/journal.pmed1000097

Fig. 1. The PRISMA flow diagram of the literature search and review process

3.3 Inclusion and exclusion criteria

In general, all the titles and abstracts of the eligible articles were read carefully before being selected based on the inclusion and exclusion criteria. Each article must fulfil all of the inclusion criteria and must not match any of the exclusion criteria. First, only journal papers with empirical data were chosen as sources of literature. The review omitted all book series, book chapters, conference proceedings and review articles. Second, the articles must be published in English, and all languages other than English were excluded. Thirdly, regarding the publication year, the period between January 2019 and January 2021 appeared to be a reasonable timeframe for the most current

research trends on the use of MALL in English teaching and learning. As the review concentrated on figuring out the recent studies conducted on the implementation of MALL in Malaysia, related articles indexed in Social Sciences, Arts and Humanities and Computer Science were selected while other articles indexed in other subject areas were excluded. Moreover, since the review aimed to highlight the current trends and practices of MALL in Malaysia, only studies involving Malaysian students or teachers in the Malaysian education system were selected. The criteria implied that studies conducted on students and teachers from other nationalities in different contexts than Malaysia were also excluded from the review (see Table 1).

Criteria	Inclusion	Exclusion
Document type	Article journal (empirical data)	All book series, book chapters, conference proceedings and review articles
Language	English	Non-English
Timeline	January 2019–January 2021	<january 2019="" and=""> January 2021</january>
Indexes	Social Sciences, Arts and Humanities and Computer Science	Not indexed in Social Sciences, Arts and Humanities, and Computer Science
Context	Malaysia	Other than Malaysia

 Table 1. Inclusion and exclusion criteria

Additionally, the review included additional criteria to narrow the search and provide more relevant results that addressed the study's research question. The additional criteria outlined for the review were that the eligible studies: (1) must be related to mobile-assisted language learning, (2) must be related to English language learning through the use of mobile devices/mobile technology, (3) must be related to ESL or EFL teachers, and (4) must be related to English language learning through the use of mobile devices/technology by mainstream education students.

3.4 Systematic review process

The systematic review process entailed the identification of keywords, screening of records, determining the eligibility of records, and finally, the inclusion of studies. Figure 1 illustrates the process of identifying and selecting articles for this systematic review of the literature.

Identification. Identification is the first stage of the systematic review process. The authors began the identification process by generating a range of keywords derived from the research question, thesaurus and prior relevant studies relative to MALL. The three databases were combed with these exact keywords: "mobile-assisted language learning", "mobile-assisted language learning", "mobile technology", "mobile device*", "smartphone*", "mobile phone*", "English', "English language learning", "ELT", "ESL", and "second language learning" (see Table 2).

The search yielded 4027 articles from three databases: 780 from Elsevier's Scopus, 2864 from Taylor & Francis, and 383 from ERIC. Despite the authors' decision to utilise a 2-year timeline to determine the trends in the usage of MALL activities in Malaysia

and using this inclusion criterion (publication year) at the start of the searching procedure, the search generated far too many articles from the three databases. The timeline period was established following an assertion made by [66], who mentioned that it is almost unrealistic for researchers to examine all published papers; hence, researchers should define the timeline period to analyse and report the review findings effectively.

Additionally, the large number of records was ascribed to the extensive use of keywords while searching for MALL publications published between January 2019 and January 2021 in Elsevier's Scopus, Taylor & Francis, and ERIC databases. Furthermore, the massive number of articles was compounded by a timeline sorting feature limitation in the ERIC database since the database only provides the "last five years" (since 2017) option. Resultantly, the articles retrieved from the ERIC database were manually screened using Microsoft Excel to remove those that were not published between January 2019 and January 2021.

	- TEL 1					
Table 7	The coerch	ofring ligod	in the	avatomotio	POLIOIT	progod
TADIC 4.	THE SEALCH	SUITINE USED		Systematic	ICVICW	01000555
		~		~ / ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~		p

Databases	Keywords Used
Elsevier by Scopus	TITLE-ABS-KEY (("mobile-assisted language learning" OR "mobile assisted language learning" OR "m-learning" OR "mobile learning" OR "mobile technology" OR "mobile device*" OR "smartphone*" OR "mobile phone*") AND ("English" OR "English language learning" OR "ELT" OR "ESL" OR "second language learning"))
Taylor & Francis	[[All: "mobile-assisted language learning"] OR [All: "mobile assisted language learning"] OR [All: "m-learning"] OR [All: "mobile learning"] OR [All: "mobile technology"] OR [All: "mobile device*"] OR [All: "smartphone*"] OR [All: "mobile phone*"]] AND [[All: "english"] OR [All: "english language learning"] OR [All: "elt"] OR [All: "elt"] OR [All: "escond language learning"]] AND [Publication Date: (01/01/2019 TO 12/31/2021)]
ERIC	(("mobile-assisted language learning" OR "mobile assisted language learning" OR "m-learning" OR "mobile learning" OR "mobile technology" OR "mobile device*" OR "smartphone*" OR "mobile phone*") AND ("English" OR "English language learning" OR "ELT" OR "ESL" OR "second language learning"))

Screening. The second stage was the screening procedure, in which the authors automated the screening of all 4027 articles by utilising the sorting features available in the three databases. Besides, the authors also applied the inclusion and exclusion parameters determined at the beginning of the review process to determine the articles that best addressed the research question (see Table 1). At this stage, 3,984 of 4,027 articles were eliminated from the screening process because (1) they were not articles with empirical data, (2) they were not written in English, (3) they were indexed in other subject areas other than the indexes mentioned in the inclusion criteria, such as Business, Management and Accounting Index, Economics, Econometrics and Finance Index, and Engineering Index, and (4) they were not conducted within the context of Malaysia's educational system and included non-Malaysian participants (teachers and students). The remaining 43 articles were subsequently included in the third phase of the systematic review process, the eligibility phase.

Eligibility. During the eligibility phase, the authors carefully analysed and filtered the remaining 43 articles to verify that each article met all of the additional criteria

established by the authors at the outset of the study in order to answer the research question (see Table 1). This was achieved by reading the 43 full-text publications systematically rather than just skimming the titles and abstracts. This procedure eliminated 32 articles that did not meet the authors' additional criteria, since they were: (1) online learning studies with no connection to mobile learning, (2) mobile learning studies with no relation to English language learning, (3) focused on general teachers who were not necessarily ESL or EFL teachers, and (4) studies of English language learning with the use of mobile devices/technologies by non-mainstream education students (special education and students with disabilities). After this procedure concluded, only eleven eligible articles remained.

Inclusion. The remaining eleven articles were appraised for quality at the culmination of the systematic review process, the Inclusion phase. This was achieved by submitting the articles for quality assessment to two experts. Following that, the articles were appraised using a checklist established to ascertain the trustworthiness of the studies. The checklist included the following:

Are the studies' objectives well-defined? Is the methodology for the study well-articulated? Are the findings of the study plausible and backed by evidence?

Experts and authors must concur that the articles adhered to the checklist's guidelines to be eligible for inclusion in the review. Additionally, any points of contention were discussed prior to including or excluding articles from the review. Following this, all eleven remaining articles were determined to be eligible for the review and were analysed using both quantitative and qualitative techniques.

3.5 Data abstraction and analysis

To address the study's research question of providing an up-to-date synthesis of MALL research trends in the Malaysian context, the eleven articles that fulfilled the inclusion criteria were analysed and synthesised using qualitative and quantitative techniques. Both qualitative and quantitative techniques to data synthesis enable researchers to "generate a robust, empirically derived answer to a focused research question" [67, p. 446]. Based on the research question, seven codes were used for the qualitative analysis: (1) research objectives, (2) research methodologies, (3) educational level, (4) learning context, (5) device types, (6) theories, and (7) research outcomes. The codes were obtained from systematic reviews of the literature conducted by [68] and [69] with the primary objective of gaining insight into research trends and practices concerning the use of mobile technologies in education. The two experts that helped with article assessment during the Inclusion phase assisted with the code verification process as well. The experts were engaged in both phases since they were not only competent in qualitative analysis, but also because their reading of the articles during the Inclusion phase enriched their perspectives and judgement about code selection. Accordingly, both experts concurred that the codes used for this review were pertinent to the study's single overarching research question.

The writers analysed the eleven papers in a five-step approach. First, the authors combed through all articles to formulate an answer to the study's research question. Next, the authors highlighted and tagged the information from each of the eleven studies and mapped the information to the appropriate codes derived from the study's primary research question. The authors then abstracted the data and entered it into a Microsoft Excel spreadsheet for subsequent quantitative and qualitative analysis. Following that, the authors synthesised the qualitative data before conducting quantitative data analysis, using frequency data analysis to determine the number of occurrences for each code. Finally, a comprehensive frequency table was created. The frequency table enables readers to instantly detect trends in a corresponding data set by providing a snapshot of the data in a concise manner. All of the aforementioned procedures were cross-checked by the two authors to ensure the review's accountability and robustness.

4 Findings and discussion

This section presents the findings and discussion in reference to seven codes pertinent to the research question of the present systematic literature review. The codes are (1) research objectives, (2) research methodologies, (3) educational level, (4) learning context, (5) device types, (6) theories, and (7) research outcomes.

4.1 Background of the selected articles

The review identified eleven papers that offered critical information concerning research trends and MALL practices in Malaysia. Seven codes adopted from two previous and established systematic reviews of the literature were analysed and synthesised. The codes were research purposes, methodologies, educational level, learning setting, device types, theories, and research outcomes. Out of eleven articles, seven were published in 2019, and four were published in 2020. In addition, four of the eleven studies were conducted in Selangor and one in each of the other states, namely Negeri Sembilan, Sarawak, Terengganu, and Kelantan. Three studies omitted information on the study's location.

4.2 Research purposes

As indicated in Table 3, the most often documented research purpose was to determine the effects of MALL activities on students' English performance, as five out of eleven studies focused only on this topic. The performance of students was evaluated in several aspects related to language skills. For instance, [71] investigated the use of social media in helping students improve their spelling skills. [28] investigated the use of WhatsApp to help students develop their narrative writing skills. Meanwhile, [72] examined the use of mobile self-access learning through Quizziz to assist students in acquiring idioms and improving their writing. [73] investigated the effects of using mobile-assisted vocabulary learning on students' vocabulary development. [74]

assessed students' performance in English class following the introduction of guided mobile learning intervention.

The studies on the effects of MALL were the prime focus of current research trends and practices in Malaysia since mobile learning is a growing phenomenon in the country, and the usage of mobile devices in the educational system is intended to expand and enhance students' learning [18], [75]. Additionally, the high level of interest in conducting studies on the effects of MALL implementation in the Malaysian context indicates that mobile technologies are rapidly attracting researchers and educators in Malaysia to address the increasing demand for a robust understanding as to how mobile devices may be used to enhance English teaching and learning [76], [77].

The second most-often occurring research purpose, accounting for three out of eleven studies, was to assess students' perceptions and experiences with MALL activities in the context of English learning. Students' perceptions and experiences with MALL activities were evaluated in a multitude of aspects, including their perceptions about the usage of mobile learning to enhance MUET students' writing proficiency [43], their perceptions toward the use of MP3 in non-formal listening activities [78], and their perceptions regarding the use of technology in secondary school English instructions in Malaysia during the Covid-19 pandemic [79].

Studying students' perceptions and experiences was the second major trend in MALL research and practice in Malaysia, as an in-depth understanding of how each student experiences learning as they come from diverse backgrounds is critical in ensuring the successful implementation of mobile learning [45]. Additionally, [68] emphasised the need to research students' perceptions and experiences since students' perceptions of the usefulness of MALL activities may influence their propensity to adopt the learning practice. Understanding students' perceptions and experiences are critical because input from students who are also technology users will assist researchers and educators in developing successful MALL practices that will ultimately become a powerful tool in teaching and learning English.

From January 2019 to January 2021, a study undertaken by [80] was the only study that ascertained students' acceptance of the usage of MALL in learning English vocabulary and the expected app features. It is alarming that no more research is being undertaken in this domain. This situation must change since advocating for the use of technology in education will be challenging without students' acceptance and comprehension of the technology's usefulness [81]. Furthermore, it is essential to comprehend the variables that contribute to students' acceptance of mobile learning in the context of English learning, as students' acceptance of mobile technologies has a significant impact on their future intention to use MALL [80] as well as on their academic achievement [82].

Furthermore, [83] performed the only research that examined the effect of gender as a moderator in the relationships between variables in the Unified Theory of Acceptance and the Use of Technology (UTAUT) Model and students' attitudes regarding the use of m-learning in English learning. Similar to the acceptance study, this was the only study that examined the moderating impact of gender and students' attitudes towards MALL usage. Additional studies should be conducted to add to the existing body of knowledge regarding the variables that affect students' attitudes toward MALL practices and the moderating effect of gender on the relationships between the specified variables in UTAUT. Assessing the variables affecting students' attitudes is necessary because "mobile learning does not happen in a vacuum" [68, p. 58], and effective implementation of MALL demands a comprehensive knowledge of the key variables influencing students' attitudes concerning MALL [84], [85]. Furthermore, it is critical to assess students' attitudes since positive student attitudes encourage mobile learning, while negative student attitudes may result in a lack of usage [86].

Besides, further study into the moderating effect of gender is needed to provide robust empirical evidence on the role of gender in affecting students' attitudes towards MALL. According to [87], while several previous studies have demonstrated that gender has no effect on students' intentions to adopt new technology in specific settings, there are instances where gender has been demonstrated to be a deciding factor in students' willingness to accept technologies in learning. Furthermore, prior research also indicated that the effects of gender on the use of technology for learning vary according to the study's subject matter and cultural background [88]. Thus, acknowledging the gender dimension in MALL in Malaysia is critical for informing researchers and educators about the factors to consider when designing MALL activities that appeal to both male and female students.

Finally, only one of eleven studies examined students' smartphone usage behaviours. [89] investigated how distance education students utilise their smartphones for academic purposes, namely to complete their English coursework. The research of students' behaviour while using smartphones for educational purposes in Malaysia is critical, yet limited. According to [90] and [91], empirical data on smartphone usage behaviours is critical because it may provide answers to the ongoing debate over the benefits and drawbacks of incorporating mobile phones into educational activities. It is helpful to understand students' smartphone behaviour for smartphones to evolve into more than merely a communication tool and instead into a valuable educational tool that supports sustainable learning [89], [92]. By gaining an understanding of how students use mobile phones for learning, researchers and educators may develop mobile-friendly MALL strategies which are reflective of students' behavioural usage of smartphones; thereby accommodating students' diverse learning styles [89], [93].

4.3 Research methodologies

The systematic review of the literature indicated that eleven studies published between January 2019 and January 2021 used three different types of research methodologies. The quantitative research method was the most often used method to investigate contemporary MALL trends and practices in Malaysia, accounting for six studies, followed by four studies that employed a mixed-method approach. The qualitative method was the least often used, with just one study (see Table 3).

From six of the quantitative studies, three studies used a survey method with questionnaires, two used a quasi-experimental design with pretest-posttest control groups, and only one study used quantitative action research of three levels to elicit empirical data about Malaysian students' experiences, perceptions, and effects of MALL implementation and intervention.

The review findings support the assertions made by [94] and [95] that researchers in the field of education are more inclined to conduct quantitative research. This scenario

may be attributable to the Malaysian researchers' preference for quantitative study over qualitative study to understand the burgeoning phenomena of MALL application in supporting English language learning. As emphasised by [96] and [97], quantitative studies enable researchers to describe trends and relationships between study variables using statistical analysis, which is more reliable, objective, and accurate using established computational techniques that support research generalisations.

Mixed-method approach was the second most often employed research methodology, accounting for four of the eleven studies. Two studies integrated questionnaires and semi-structured interviews; one study incorporated questionnaires and participant's self-report, and one study combined questionnaires and open-ended questions. The use of a mixed-method approach highlights the vital importance of investigating and interpreting MALL activities in Malaysia from a variety of perspectives. The findings of the SLR correspond to several academics' statements on the increasing relevance of the mixed-method approach in social sciences research over the last several decades [98], [99].

The increased relevance of mixed-method research in the education field stems from the research design's central objective and principle, which is to integrate quantitative and qualitative approaches in order to obtain a more comprehensive understanding of research issues and complicated phenomena than each approach alone could offer [100], [101]. Furthermore, the mixed-method approach is praiseworthy because it allows researchers to acquire both the depth and breadth [100] of information regarding MALL practices in Malaysia by triangulating one set of findings with another, thus improving the validity and completeness of the evidence [102].

Between January 2019 and January 2021, the qualitative method was the least often employed method in conducting research in Malaysia, accounting for just one of eleven studies conducted during that time period. The study performed by [28] used a case study research design to explore the effects of MALL implementation among students in Malaysia. As shown by the dearth of qualitative research in Malaysia, it appears that researchers are more interested in objectively comprehending the MALL phenomenon than they are in researching it in profundity.

This scenario is certainly a result of common criticisms levelled against qualitative research, including its small sample size, anecdotal character, lack of rigour, reliability and validity concerns, inability to be extrapolated, uncontrolled bias, and subjectivities [103], [104], [105]. While qualitative data provide well-founded, thorough, and extensive descriptions and explanations of participants' personal views on the phenomena under investigation, analysing qualitative data is often an unexpectedly tricky, iterative, complicated, ambiguous, and time-consuming process [106], [107], [108]. Even with established qualitative data analysis software, researchers generally struggle to elucidate the implicit meanings that people associate with their actions and responses to a phenomenon without losing sight of their assumptions, opinions, and personal beliefs that may influence the analysis [106], [107], [109]. As a result of the external criticism and analytical difficulty associated with qualitative research, many researchers continue to avoid it even though more qualitative research is needed in the MALL field in Malaysia because of its diversity, quality, and authority that can aid in the development of a more nuanced understanding of educational problems.

				1									· · · · · ·	,
			NB										<u> </u>	<u> </u>
	rch nes		NEC											
	sear tcon		¥Т			~								
	Re Ou	POS	РЕВ	<u> </u>	~						~			
			ПS				-	~	-	~		/		
	تە تە		٧N	<u> </u>	~		~		~		~	/	~	
	Theorie		V			/ (TAM)		/(ZPD)		/ (DCE)				/ (UTAUT)
	of	s	IN		~				<u> </u>				~	~
	pes	evic	TS				~							
	Ê		dS	~		~		~		~	~	~		
	u		SN			~							~	
	Icati	ure	NE		<u> </u>		<u> </u>				<u> </u>			
	Edu	5	11						<u> </u>					
			<u>в</u>											
	onal		НЕ		~	<u> </u>					_	`	<u> </u>	<u> </u>
s	ıcati	Leve	SS	~					~	<u> </u>				
ding	Edu		Sd				-	~						
le fin		QL	SD					~						
3. TF			0+00	<u> </u>										
ble	ogy	MM	48 +9		~									
Ta	lobo		ISS +Ò			~					~			
	Meth		алд						~					
		NO	бe							~		~		
			ð				~						~	~
	<u> </u>		NB								~			
	ses		СA											~
	rpo		VS			~								
	Pu		bE	<u> </u>	<u> </u>								<u> </u>	
			EI				<u> </u>	<u> </u>	<u> </u>	<u> </u>		<u> </u>		
		I acality	Locality	Sarawak	Terengganu	Not specified	Selangor	Negeri Sembilan	Selangor	Selangor	Not specified	Kelantan	Selangor	Not specified
		Voor	Тсаг	2019	2019	2019	2019	2019	2019	2019	2020	2020	2020	2020
		Studios	samme	John and Md. Yunus	Ahmad Tajuddin et al.	Deris and Shukor	Lee et al.	Suhaimi et al.	Abdullah et al.	Govindasamy et al.	Annamalai and Kumar	Siew et al.	Azar and Iskandar Tan	Hashim et al.

Purposes	Methodology	Educational Level	Educational Context	Types of Devices	Theories	Research Outcomes
EI= Effects of	QN= Quantitative	PS= Primary	F= Formal	SP= Smartphones	A= Available	POS= Positive
integration	CS= Case study	school	IF= Informal	ST= Smartphones and tablets	NA= Not available	NEG= Negative
PE= Perceptions and	MM = Mixed method	SS=	NF= Non-formal	NI= Not indicated	TAM= Technology	NR= Not related
experiences	QL= Qualitative	Secondary	NS= Not specified		Acceptance Model	SL= Students'
SA= Students'	Q= Questionnaire	school			ZPD= Zone of Proximal	learning
acceptance	QE= Quasi experiment	HE= Higher			Development	PER = Perceptions
GA= Moderating role	QAR= Quantitative action	education			DCE= Dale's Cone of	and experiences
of gender and	research				Experience	TA = Technology
students' attitudes	Q+SSI= Questionnaire and				UTAUT= Unified Theory	acceptance
UB= Students'	Semi-structured interview				of Acceptance and	
smartphone usage	Q+SR= Questionnaire and				Technology Use	
behaviour	Participants' self-reports					
	Q+OQ= Questionnaire and					
	Open-ended questions					

4.4 Educational level

Table 3 summarises the educational level reported in the eleven research articles included in the systematic review. Higher education institutions (HEIs) level was the most researched educational level, accounting for six of eleven studies and secondary school was the second most researched educational level, accounting for three studies. On the other side, primary school was the least studied educational level, accounting for just two of the eleven studies. The data indicates that the researchers were nearly twice as likely to investigate students in HEIs than they were in elementary and secondary schools. Previous systematic reviews of the literature reported similar findings, namely that the majority of researchers were more inclined to study MALL practices among students enrolled at HEIs than at other educational levels [10], [110], [111].

The findings of the current SLR demonstrate that more studies were undertaken at institutions of higher education than at the lower educational levels. This trend is explained by the widespread use of m-learning in HEIs, since older students and HEIs are perceived to be better equipped for and adaptable to the transition to Education 4.0 [112], [113]. The findings echo those of [111], who found that primary school is the least studied educational level in MALL research. The scarcity of studies examining MALL integration in the context of primary school students in Malaysia is because mobile learning has not been incorporated formally within the framework of primary school instruction due to the policy prohibiting primary school students from using mobile devices in school [80], [114], [115].

4.5 Learning context

The learning context is the environment in which something is learned or comprehended. [116] identified three distinct dimensions of the learning context: (1) formal context, which refers to learning meant to occur in a formal classroom environment; (2) informal context, which refers to learning intended to occur outside the classroom; and (3) non-formal context, which refers to learning that is not intended to occur in a formal or informal classroom setting.

Since the majority of the eleven studies included in the systematic review did not specify the learning contexts in which the research was conducted, the learning contexts for the majority of the studies were inferred through reading the articles using classification description by [116]. However, when [78] investigated the usage of non-formal MP3 listening activities, they understood non-formal context differently than [116]. [78] defined non-formal contexts using the Luxembourg Ministry of Family and Integration's (2013) definition, which identifies "non-formal education as organised educational activity outside of the classroom that is intended to serve identifiable learning groups with identifiable learning objectives" [78, p. 86]. However, as described by [116], the non-formal environment emphasises the unplanned learning that happens in both formal and informal classroom settings. Thus, for the purpose of this systematic review and in line with the definition of learning environment established by [116], the learning context for the study carried by [78] was classified as informal rather than non-formal. This is because the listening activities used in the research by [78] were prearranged and were designed to aid students in improving listening skills outside of the classroom.

The proportion of learning context recorded in eleven studies is shown in Table 3. The SLR analysis revealed that the formal learning context was the most prevalent, accounting for four studies, followed by the informal learning context, accounting for three studies and the non-formal learning context accounted for two studies. Two studies, on the other hand, did not specify the context in which MALL practices and research occurred. While mobile learning allows for greater flexibility in terms of time and location, most studies describing the advancements in English learning experiences for secondary school and higher education students took place in physical classrooms or on school and university premises.

Numerous learning contexts were utilised to integrate MALL for students at HEIs, indicating the breadth of efforts to globalise learning and equip students with the technological skills necessary to enter the twenty-first century workforce while also adhering to the mandate outlined in the Malaysia Education Blueprint (MEB) 2013–2025 [117]. The MEB 2013–2025 requires Malaysian HEIs to incorporate blended learning in up to 70% of their programmes, which is the generic pedagogical approach at the tertiary level, by including a broad range of technology-based learning opportunities through mobile applications, YouTube videos, internet sites, education websites, and virtual meeting and conferencing [117], [118]. This situation has enabled educators and researchers at Malaysian HEIs to examine the effects of mobile technology on students' English acquisition across a range of contexts (formal, informal, and non-formal) to understand better the differences between learning environments and their affordances.

Interestingly, despite a policy banning secondary school students from using smartphones, innovations in English instruction were most prevalent in the formal learning context, followed by the informal learning context for upper form students (Form 5 and pre-university students) in Malaysia. This scenario could be because the school authorities involved in the studies permitted upper-form students to bring mobile phones to school, allowing teachers to plan formal and informal MALL activities to supplement English instruction without relying on non-formal activities that were more difficult to plan and execute. Nevertheless, due to the current policy prohibiting the use of mobile phones on school grounds [80], [114], [115], [119], [120], MALL integration at the primary school level was conducted exclusively outside of the classroom setting, with both intentional and unintentional learning activities aimed at enhancing students' English language acquisition.

4.6 Types of devices

Table 3 depicts the different types of mobile devices mentioned in the articles chosen for this systematic review. Six of the eleven studies reported utilising smartphones as their primary device for MALL activities, and one study reported using both smartphones and tablets for MALL integration. Meanwhile, the other four studies did not specify the mobile devices utilised for MALL activities.

As shown in Table 3, smartphones were the most often used mobile device for MALL integration in the Malaysian educational context, as demonstrated by six studies that employed smartphones as the major instrument for MALL practices. This conclusion is similar with the findings of a systematic review of the literature undertaken by [111], who discovered that smartphones were the most commonly utilised type of mobile

device during MALL activities. The pervasive use of smartphones in MALL activities may be explained by the high rate of smartphone ownership among Malaysian students, as reported by [5], who highlighted that 86.3% of the respondents under the age of 20 owned a smartphone, including those who are possibly students in Malaysia's school system. Additionally, [113] claimed that students own smartphones irrespective of their demographic characteristics, which may explain why smartphones were the most often used mobile device for MALL activities in Malaysia. Furthermore, a meta-analysis by [10] reported that both smart and non-smart portable mobile devices were widely used and outperformed larger-screen mobile devices in language learning facilitation. This is because handheld devices are more compact, enabling educators and students to do MALL activities in various learning settings, whether formal, informal, or non-formal [10].

4.7 Theories

The systematic review examined the most frequently used theoretical frameworks in MALL research conducted in Malaysia between January 2019 and January 2021. In general, four out of eleven studies were structured around appropriate theoretical frameworks, while seven studies did not specify the theory or pedagogical method utilised in their MALL investigations (see Table 3). The findings of the current SLR corroborates those of [69], who observed that only a few studies performed between 2000 and 2012 were framed around relevant theories.

Two of the studies included in this SLR used acknowledged frameworks for technology adoption and acceptance, including the Technology Acceptance Model (TAM) (1) and the Unified Theory of Acceptance and Technology Use (UTAUT) (1). The TAM and the UTAUT models have been widely, successfully, and appropriately used in information technology (IT) and information system (IS) disciplines to explain, predict, and comprehend individual acceptance and use of technology [121], [122]. [123] also reported similar findings in a comprehensive review of the literature on mobile learning in Saudi Arabia, noting that the TAM and UTAUT frameworks were the most often used frameworks in research on technology adoption and acceptance.

On the other hand, the other two studies used frameworks based on learning theories such as constructivism (1) and Dale's cone of experience, also known as experiential learning (1). The findings support the observations by [69, p. 7] where the learning theories and models studied in MALL research "often originated from grand theories of learning, including constructivism, social constructivism, socio-cultural theory, and situated learning theory".

4.8 Research outcomes

Eleven research papers were analysed in this review. Nevertheless, only nine of the eleven studies were evaluated for their research outcomes since they offered evidence of MALL efficacy in three domains: students' learning performance, perceptions, and technology acceptance. Five studies examined the effect of mobile devices on students' English performance and achievement, three studies examined students' perceptions

and experiences with MALL practices used to enhance and support English acquisition, and one study examined students' acceptance of MALL activities used to learn English. The outcomes were categorised using a framework provided by [68] for categorising research outcomes, namely positive, negative, and neutral.

Each of the nine studies reported positive outcomes. This scenario revealed that every research conducted in Malaysia to establish the efficacy of MALL concluded that utilising mobile devices improved students' learning performance, perceptions, and experiences, as well as their acceptance of MALL activities during the English learning process. Interestingly, when reviewing studies on the use of mobile learning in science education, a study performed by [124] also discovered that regardless of the study's objective, all 49 studies included in their review exhibited positive learning outcomes. The consistency of outcomes across several disciplines of education suggests that mobile technologies, irrespective of subject matter, are adaptive and beneficial for educational application.

Nevertheless, the unanimous positive research outcomes suggest that researchers are less likely to publish studies with negative outcomes than those with positive outcomes. According to [125] and [126], many researchers believe that negative outcomes are statistically insignificant and may reflect poorly on their credibility and academic credentials, leading them to avoid publishing the aforementioned studies. Additionally, this could be due to the penchant of journals and publications in publishing positive findings, which results in publication bias [127], [128]. Publication bias has various effects on the corpus of scientific knowledge, including skewing it towards statistically significant or "positive" outcomes [127], [128].

5 Identified gaps and recommendations for future studies

This systematic review has identified eight research gaps. First, between January 2019 and January 2021, only eleven research performed in Malaysia were published in peer-reviewed journals on three major electronic databases. Furthermore, the eleven studies were undertaken only in Sarawak, as well as in Peninsular Malaysia's east coast, central, and southern regions, namely Kelantan, Terengganu, Selangor, and Negeri Sembilan. As a result, further studies in the MALL discipline are required, especially in the areas of English language teaching in northern Peninsular Malaysia and Sabah, in order to offer a holistic representation of MALL application in English language education in the Malaysian context.

Second, the findings indicated that most MALL research in Malaysia focused on investigating the effects of using mobile devices in the teaching and learning of English. Future research should focus on other purposes, including students' perceptions and attitudes toward MALL, factors that influence Malaysian students' acceptance and use of mobile devices in English learning, behavioural use of mobile devices by Malaysian students, and specific mobile apps that can aid students' English acquisition.

Third, this review uncovered a scarcity of qualitative research on MALL in Malaysia. Future research on MALL in Malaysia should adopt the qualitative research method in order to provide the academic community with in-depth insights and perspectives on the phenomena being examined that quantitative research cannot provide [129].

Fourth, most of the studies were conducted in Malaysian institutions of higher learning. Thus, further research at the secondary and primary school levels in Malaysia is necessary to have a better understanding of the impacts of mobile devices on diverse groups of students throughout the Malaysian education systems, especially in English language education. This proposal is concordant with the recommendations from two of the articles reviewed; [80] and [89] recommended that future research should involve participants from a range of educational institutions in order to improve the heterogeneity of MALL studies.

Fifth, as previously mentioned, the majority of studies in Malaysia were undertaken in a formal learning context for secondary school and higher education students, but none for primary school students. Future research on MALL integration in Malaysian secondary schools and higher education institutions should concentrate on the use of mobile devices in informal and non-formal learning contexts. In contrast, research at the primary school level should focus on the usage of mobile devices in the formal learning contexts to better understand the impact that different learning contexts may have on the efficacy of MALL integration.

Sixth, since seven of the eleven studies lacked a theoretical framework, future research on MALL practices in Malaysia must exhibit academic rigour and the capacity to substantiate provided claims via the inclusion of relevant theories. Without a theoretical perspective, studies will have an unclear framework and vision that suggest methodological flaws [69], [130]. Additionally, the theoretical framework strengthens the academic and educational character of the study's arguments by enabling researchers to add depth to the data analysis by referencing theories presented by subject experts that are often already established or indisputable [131].

Seventh, nine of the studies included in this review had positive outcomes. Future research on the impacts of MALL integration in English language classes in Malaysia should concentrate on recognising intervening variables that may account for the observed positive outcomes. Furthermore, future studies are expected to be transparent in reporting negative outcomes to assist other academics, educators, and stakeholders in averting repeated failed efforts. Finally, teachers' perspectives were mostly absent from this review. It would be beneficial to learn about Malaysian teachers' views and concerns since this would provide valuable data on an essential element of implementing MALL practices in Malaysian classrooms.

6 Conclusion

The present systematic literature review has accordingly addressed the study's overarching research question concerning current MALL research trends and practices in Malaysia by reviewing eleven articles published between January 2019 and January 2021. The findings of this review have provided an up-to-date synthesis of MALL research in Malaysia's English language education system in terms of research purposes, methodology, educational level, learning context, device types, theories, and research outcomes. Despite mobile technology's immense growth and potential, the review has discovered that research on MALL and its implementation in English language instruction in Malaysia's education system is still in its formative stages. Thus,

substantial studies are required to address the gaps identified in the current review so as to establish effective and practical strategies for MALL implementation across all educational levels in Malaysia's English language education system in compliance with Education 4.0 requirements.

7 References

- R. Kaliisa, E. Palmer and J. Miller, "Mobile learning in higher education: A comparative analysis of developed and developing country contexts", *British Journal of Educational Technology*, vol. 50, no. 2, pp. 546–561, 2017. Available: <u>https://doi.org/10.1111/bjet.12583</u>
- [2] H. K. Lamptey and R. Boateng, "Mobile learning in developing countries: A synthesis of the past to define the future.", *International Journal of Educational and Pedagogical Sciences*, vol. 11, no. 2, pp. 448–455, 2017. Available: <u>https://publications.waset.org/10006532/</u> mobile-learning-in-developing-countries-a-synthesis-of-the-past-to-define-the-future
- [3] N. M. Nordin, M. A. Embi, H. Norman, and E. Panah, "A historical review of mobile learning research in Malaysia and its implications for Malaysia and the Asia-Pacific region," in *Mobile Learning in Higher Education in The Asia-Pacific Region*, A. Murphy, H. Farley, L. Dyson, and H. Jones, Eds. Singapore: Springer, 2017, pp. 137–150. <u>https://doi. org/10.1007/978-981-10-4944-6_7</u>
- [4] J. Burston, "A selected annotated bibliography of implementation studies 1994–2012", Language Learning & Technology, vol. 17, no. 3, pp. 157–225, 2013. Available: <u>http://llt.msu.edu/issues/october2013/burston.pdf</u>
- [5] J. Müller, "Malaysia: Smartphone ownership by age 2019 | Statista", *Statista*, 2021.
 [Online]. Available: <u>https://www.statista.com/statistics/973694/malaysia-smartphone-ownership-by-age/</u>. [Accessed: 05- Apr- 2021].
- [6] S. Shahrol, S. Sulaiman, M. Samingan and H. Mohamed, "A systematic literature review on teaching and learning english using mobile technology", *International Journal of Information and Education Technology*, vol. 10, no. 9, pp. 709–714, 2020. Available: <u>https://doi.org/10.18178/ijiet.2020.10.9.1447</u>
- [7] S. Papadakis, F. Alexandraki and N. Zaranis, "Mobile device use among preschool-aged children in Greece", *Education and Information Technologies*, pp. 1–34, 2021. Available: <u>https://doi.org/10.1007/s10639-021-10718-6</u>
- [8] S. Papadakis, "Advances in mobile learning educational research (A.M.L.E.R.): Mobile learning as an educational reform", *Advances in Mobile Learning Educational Research*, vol. 1, no. 1, pp. 1–4, 2021. Available: <u>https://doi.org/10.25082/AMLER.2021.01.001</u>
- [9] H. Crompton, D. Burke and K. Gregory, "The use of mobile learning in PK-12 education: A systematic review", *Computers & Education*, vol. 110, pp. 51–63, 2017. Available: <u>https://doi.org/10.1016/j.compedu.2017.03.013</u>
- [10] Z. Chen, W. Chen, J. Jia and H. An, "The effects of using mobile devices on language learning: A meta-analysis", *Educational Technology Research and Development*, vol. 68, no. 4, pp. 1769–1789, 2020. Available: <u>https://doi.org/10.1007/s11423-020-09801-5</u>
- [11] J. Traxler, "Defining mobile learning", *Iadisportal.org*, 2005. [Online]. Available: <u>http://www.iadisportal.org/digital-library/defining-mobile-learning</u>. [Accessed: 12- May- 2021].
- [12] A. Kukulska-Hulme, "Will mobile learning change language learning?", *ReCALL*, vol. 21, no. 2, pp. 157–165, 2009. Available: <u>https://doi.org/10.1017/S0958344009000202</u>
- [13] V. Lee, L. Fischback and R. Cain, "A wearables-based approach to detect and identify momentary engagement in afterschool Makerspace programs", *Contemporary Educational Psychology*, vol. 59, p. 101789, 2019. Available: <u>https://doi.org/10.1016/j.cedpsych.</u> 2019.101789

- [14] M. Elaish, L. Shuib, N. Ghani and E. Yadegaridehkordi, "Mobile English Language Learning (MELL): A literature review", *Educational Review*, vol. 71, no. 2, pp. 257–276, 2017. Available: <u>https://doi.org/10.1080/00131911.2017.1382445</u>
- [15] G. Krull and J. Duart, "Research trends in mobile learning in higher education: A systematic review of articles (2011–2015)", *The International Review of Research in Open and Distributed Learning*, vol. 18, no. 7, 2017. Available: <u>https://doi.org/10.19173/irrodl.v18i7.2893</u>
- [16] W. Wu, Y. Jim Wu, C. Chen, H. Kao, C. Lin and S. Huang, "Review of trends from mobile learning studies: A meta-analysis", *Computers & Education*, vol. 59, no. 2, pp. 817–827, 2012. Available: <u>https://doi.org/10.1016/j.compedu.2012.03.016</u>
- [17] P. Serdyukov, "Innovation in education: What works, what doesn't, and what to do about it?", *Journal of Research in Innovative Teaching & Learning*, vol. 10, no. 1, pp. 4–33, 2017. Available: <u>https://doi.org/10.1108/JRIT-10-2016-0007</u>
- [18] M. Shuib, S. Azizan and M. Ganapathy, "Mobile learning readiness among English language learners in a public university in Malaysia", *Pertanika Journal of Social Sciences* & *Humanities*, vol. 26, no. 3, pp. 1491–1504, 2018. Available: <u>http://www.pertanika.upm.</u> <u>edu.my/resources/files/Pertanika%20PAPERS/JSSH%20Vol.%2026%20(3)%20Sep.%20</u> <u>2018/16%20JSSH-1897-2016.pdf</u>
- [19] J. A. Saiful, "Mobile teacher professional development (MTPD): Delving into English teachers' beliefs in Indonesia," *The International Review of Research in Open and Distributed Learning*, vol. 21, no. 4, pp. 143–160, 2020. Available: <u>https://doi.org/10.19173/irrodl.</u> v21i4.4754
- [20] N. Sofiana and H. Mubarok, "The impact of English game-based mobile application on students' reading achievement and learning motivation," *International Journal of Instruction*, vol. 13, no. 3, pp. 247–258, 2020. Available: <u>https://doi.org/10.29333/iji.2020.13317a</u>
- [21] S. Ebadi and S. Bashir, "An exploration into EFL learners' writing skills via mobile-based DynamicAssessment," *Education and Information Technologies*, vol. 26, no. 2, pp. 1995–2016, 2020. Available: <u>https://doi.org/10.1007/s10639-020-10348-4</u>
- [22] S. Jalili, H. Khalaji, and H. Ahmadi, "Vocabulary learning in the mobile-assisted flipped classroom in an Iranian EFL context," *Teaching English with Technology*, vol. 20, no. 4, pp. 82–95, 2020. Available: <u>http://cejsh.icm.edu.pl/cejsh/element/bwmeta1.element.desk-light-643dc4d5-fe8f-44b0-b8d8-35dd846560db/c/ARTICLE5.pdf</u>
- [23] B. J. Lee, "Comparing factual recall of tapped vs. handwritten text," Acta Psychologica, vol. 212, p. 103221, 2021. Available: <u>https://doi.org/10.1016/j.actpsy.2020.103221</u>
- [24] T. Sato, F. Murase and T. Burden, "An empirical study on vocabulary recall and learner autonomy through mobile-assisted language learning in blended learning settings," *Calico Journal*, vol. 37, no. 3, pp. 254–276, 2020. Available: <u>https://doi.org/10.1558/cj.40436</u>
- [25] B. G. Kim and H. R. Shim, "Chunk meets image: The effects of chunking and imagery on mobile-based self-learning of English as a foreign language," *International Journal of Computer-Assisted Language Learning and Teaching (IJCALLT)*, vol. 9, no. 1, pp. 79–98, 2019. Available: <u>https://doi.org/10.4018/IJCALLT.2019010105</u>
- [26] E. D. Reynolds and B. Taylor, "Kahoot!: EFL instructors' implementation experiences and impacts on students' vocabulary knowledge," *Computer-Assisted Language Learning Electronic Journal*, vol. 21, no. 2, pp. 70–92, 2020. Available: <u>http://callej.org/journal/21-2/</u> <u>Reynolds-Taylor2020.pdf</u>
- [27] S. F. Ng, M. A. K. Azlan, A. N. A. Kamal and A. Manion, "A quasi-experiment on using guided mobile learning interventions in ESL classrooms: Time use and academic performance," *Education and Information Technologies*, vol. 25, no. 6, pp. 4699–4719, 2020. Available: <u>https://doi.org/10.1007/s10639-020-10191-7</u>
- [28] N. D. Suhaimi, M. Mohamad and H. Yamat, "The effects of WhatsApp in teaching narrative writing: A case study," *Humanities & Social Sciences Reviews*, vol. 7, no. 4, pp. 590–602, 2019. Available: <u>https://doi.org/10.18510/hssr.2019.7479</u>

- [29] B. Al-Bogami and T. Elyas, "Promoting middle school students' engagement through incorporating iPad apps in EFL/ESL classes", *SAGE Open*, vol. 10, no. 2, pp. 1–18, 2020. Available: <u>https://doi.org/10.1177/2158244020926570</u>
- [30] A. A. F. Alzubi and M. K. A. Singh, "Investigating reading learning strategies through smartphones on Saudi learners' psychological autonomy in reading context," *International Journal of Instruction*, vol. 12, no. 2, pp. 99–114, 2019. Available: <u>http://files.eric.ed.gov/</u> <u>fulltext/EJ1211053.pdf</u>; <u>https://doi.org/10.29333/iji.2019.1227a</u>
- [31] C. Cheng and C. Chen, "Investigating the impacts of using a mobile interactive English learning system on the learning achievements and learning perceptions of student with different backgrounds", *Computer Assisted Language Learning*, pp. 1–26, 2019. Available: <u>https://doi.org/10.1080/09588221.2019.1671460</u>
- [32] R. Huang, C. Yu, T. Tang and S. Chang, "A study of the use of mobile learning technology in Taiwan for language learning", *Innovations in Education and Teaching International*, vol. 58, no. 1, pp. 59–71, 2019. Available: <u>https://doi.org/10.1080/14703297.2019.1628798</u>
- [33] G. Howlett and W. Zainee, "21st century learning skills and autonomy: Students' perceptions of mobile devices in the Thai EFL context," *Teaching English with Technology*, vol. 19, no. 1, pp. 72–85, 2019. Available: <u>http://files.eric.ed.gov/fulltext/EJ1204626.pdf</u>
- [34] P. Wongsuriya, "Improving the Thai students' ability in English pronunciation through mobile application," *Educational Research and Reviews*, vol. 15, no. 4, pp. 175–185, 2020. Available: <u>https://doi.org/10.5897/ERR2020.3904</u>
- [35] S. Şad, N. Özer, Ü. Yakar and F. Öztürk, "Mobile or hostile? Using smartphones in learning English as a foreign language," *Computer Assisted Language Learning*, pp. 1–27, 2020. Available: <u>https://doi.org/10.1080/09588221.2020.1770292</u>
- [36] S. Tümen Akyıldız and V. Çelik, "Using WhatsApp to support EFL reading comprehension skills with Turkish early secondary learners," *The Language Learning Journal*, pp. 1–17, 2021. Available: <u>https://doi.org/10.1080/09571736.2020.1865433</u>
- [37] H. R. Park, D. Kim and O. Vorobel, "International students' reading digital texts on tablets: Experiences and strategies," *Journal of Computing in Higher Education*, vol. 32, no. 3, pp. 476–504, 2020. Available: <u>https://doi.org/10.1007/s12528-019-09242-x</u>
- [38] J. F. Xin and L. A. Rachel, "Using iPads in vocabulary instruction for English language learners," *Computers in the Schools*, vol. 36, no. 1, pp. 69–82, 2019. Available: <u>https://doi.org/10.1080/07380569.2019.1565888</u>
- [39] V. N. Hoi and G. M. Mu, "Perceived teacher support and students' acceptance of mobile-assisted language learning: Evidence from Vietnamese higher education context," *British Journal of Educational Technology*, vol. e13044, 2020. Available: <u>https://doi.org/10.1111/bjet.13044</u>
- [40] T. L. N. Tran, "Perspectives and attitudes towards self-directed mall and strategies to facilitate learning for different learner groups," *CALL-EJ*, vol. 21, no. 3, pp. 41–59, 2020. Available: <u>http://callej.org/journal/21-3/Tran2020.pdf</u>
- [41] M. A. Abugohar, K. Yunus and R. A. Rashid, "Smartphone applications as a teaching technique for enhancing tertiary learners' speaking skills: Perceptions and practices," *International Journal of Emerging Technologies in Learning*, vol. 14, no. 9, pp. 74–92, 2019. Available: <u>https://online-journals.org/index.php/i-jet/article/view/10375/5666</u>; <u>https://doi.org/10.3991/ijet.v14i09.10375</u>
- [42] N. Ghorbani and S. Ebadi, "Exploring learners' grammatical development in mobile assisted language learning," *Cogent Education*, vol. 7, no. 1, pp. 1704599, 2020. Available: <u>https:// doi.org/10.1080/2331186X.2019.1704599</u>
- [43] D. S. John and M. Md. Yunus, "Students' perceptions on the use of mobile learning to improve writing proficiency in the MUET," *Pertanika Journal of Social Sciences & Humanities*, vol. 27, no. 3, pp. 1481–1492, 2019. Available: <u>http://www.myjurnal.my/public/article-download.php?id=138660</u>

- [44] R. Metruk, "EFL learners' perspectives on the use of smartphones in higher education settings in Slovakia," *Electronic Journal of e-Learning*, vol. 18, no. 6, pp. 537–549, 2020. Available: <u>https://doi.org/10.34190/JEL.18.6.006</u>
- [45] N. P. L. Nariyati, S. Sudirman and N. P. A. Pratiwi, "EFL pre-service teachers' perception toward the use of mobile assisted language learning in teaching English," *International Journal of Language Education*, vol. 4, no. 2, pp. 38–47, 2020. Available: <u>https://doi.org/10.26858/ijole.v4i2.10052</u>
- [46] T. Y. Tai, H. H. J. Chen and G. Todd, "The impact of a virtual reality app on adolescent EFL learners' vocabulary learning," *Computer Assisted Language Learning*, pp. 1–26, 2020. Available: <u>https://doi.org/10.1080/09588221.2020.1752735</u>
- [47] S. T. Saleh, "Chat and learn: Effectiveness of using WhatsApp as a pedagogical tool to enhance EFL learners 'reading and writing skills," *International Journal of English Language and Literature Studies*, vol. 8, no. 2, pp. 61–68, 2019. Available: <u>https://doi.org/10.18488/journal.23.2019.82.61.68</u>
- [48] S. Akkara, V. Anumula and M. Mallampalli, "Impact of WhatsApp interaction on improving L2 speaking skills," *International Journal of Emerging Technologies in Learning (iJET)*, vol. 15, no. 3, pp. 250–259, 2020. Available: <u>https://online-journals.org/index.php/i-jet/article/view/11534/6529; https://doi.org/10.3991/ijet.v15i03.11534</u>
- [49] Maulina, N. Noni and M. Basri, "WhatsApp audio and video chat-based in stimulating students' self-confidence and motivation to speak English," *Asian EFL Journal*, vol. 23, no. 6.3, pp. 181–203, 2019. Available: <u>https://doi.org/10.35542/osf.io/ztwbh</u>
- [50] C. Jia and K. Hew, "Supporting lower-level processes in EFL listening: The effect on learners' listening proficiency of a dictation program supported by a mobile instant messaging app", *Computer Assisted Language Learning*, pp. 1–28, 2019. Available: <u>https://doi.org/10.1080/09588221.2019.1671462</u>
- [51] M. Rashtchi and M. Mazraehno, "Exploring Iranian EFL learners' listening skills via TED talks: Does medium make a difference?," *Journal of Language and Education*, vol. 5, no. 4, pp. 81–97, 2019. Available: <u>https://doi.org/10.17323/jle.2019.9691</u>
- [52] B. Behforouz and A. Frumuselu, "The impact of text messaging as an instructional tool to enhance learner autonomy and perception," *International Journal of Learning, Teaching and Educational Research*, vol. 19, no. 11, pp. 184–202, 2020. Available: <u>https://doi.org/10.26803/ijlter.19.11.11</u>
- [53] Y. H Chang, P. R. Lin and Y. T. Lu, "Development of a Kinect-based English learning system based on integrating the ARCS model with situated learning," *Sustainability*, vol. 12, no. 5, pp. 2037, 2020. Available: <u>https://doi.org/10.3390/su12052037</u>
- [54] Z. Wang, G. J. Hwang, Z. Yin and Y. Ma, "A contribution-oriented self-directed mobile learning ecology approach to improving EFL students' vocabulary retention and second language motivation", *Journal of Educational Technology & Society*, vol. 23, no. 1, pp. 16–29, 2020. Available: <u>https://www.j-ets.net/collection/published-issues/23_1</u>
- [55] H. Karaaslan and N. Kilic, "Students' attitudes towards blended language courses: A case study", Journal of Language and Linguistic Studies, vol. 15, no. 1, pp. 174–199, 2019. Available: https://files.eric.ed.gov/fulltext/EJ1212096.pdf; https://doi.org/10.17263/jlls.547699
- [56] R. A. Yaros and J. Misak, "Completing college writing assignments on mobile phones: Comparing students' attitudes and engagement across disciplines and age", *Journalism & Mass Communication Educator*, vol. 76, no. 2, pp. 216–227, 2021. Available: <u>https://doi.org/10.1177/1077695820942422</u>
- [57] F. S. S. Alfallaj and F. Alfallaj, "Technology in Saudi EFL undergraduate classrooms: Learning tool or weapon of distraction", *Asian ESP Journal*, vol. 16, no. 4, pp. 97–115, 2020. Available: <u>https://www.asian-esp-journal.com/volume-16-issue-4-july-2020/</u>

- [58] M. H. Ko, "Students' reactions to using smartphones and social media for vocabulary feedback", *Computer Assisted Language Learning*, vol. 32, no. 8, pp. 920–944, 2019. Available: <u>https://doi.org/10.1080/09588221.2018.1541360</u>
- [59] Y. Xiao and M. Watson, "Guidance on conducting a systematic literature review", *Journal of Planning Education and Research*, vol. 39, no. 1, pp. 93–112, 2019. Available: <u>https://doi.org/10.1177/0739456X17723971</u>
- [60] Z. Munn, M. D. Peters, C. Stern, C. Tufanaru, A. McArthur and E. Aromataris, "Systematic review or scoping review? Guidance for authors when choosing between a systematic or scoping review approach", *BMC Medical Research Methodology*, vol. 18, no. 143, pp. 1–7, 2018. Available: https://doi.org/10.1186/s12874-018-0611-x
- [61] E. Van Laar, A. J. Van Deursen, J. A. Van Dijk and J. De Haan, "The relation between 21st-century skills and digital skills: A systematic literature review", *Computers in Human Behavior*, vol. 72, pp. 577–588, 2017. Available: <u>https://doi.org/10.1016/j.chb.2017.03.010</u>
- [62] M. Page et al., "The PRISMA 2020 statement: an updated guideline for reporting systematic reviews", *BMJ*, vol. 372, p. n71, 2021. Available: <u>https://doi.org/10.1136/bmj.n71</u>
- [63] D. Moher, A. Liberati, J. Tetzlaff and D. Altman, "Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement", *BMJ*, vol. 339, no. 211, pp. b2535–b2535, 2009. Available: <u>https://doi.org/10.1136/bmj.b2535</u>
- [64] J. Tennant, "Web of Science and Scopus are not global databases of knowledge", European Science Editing, vol. 46, 2020. Available: <u>https://doi.org/10.3897/ese.2020.e51987</u>
- [65] L. A. Arthurs and B. Z. Kreager, "An integrative review of in-class activities that enable active learning in college science classroom settings. *International Journal of Science Education*, vol. 39, no. 15, pp. 2073–2091, 2017. Available: <u>https://doi.org/10.1080/09500693.</u> 2017.1363925
- [66] C. Okoli, "A guide to conducting a standalone systematic literature review", Communications of the Association for Information Systems, vol. 37, no. 1, pp. 43, 2015. Available: <u>https://doi.org/10.17705/1CAIS.03743</u>
- [67] R. Mallett, J. Hagen-Zanker, R. Slater and M. Duvendack, "The benefits and challenges of using systematic reviews in international development research", *Journal of Development Effectiveness*, vol. 4, no. 3, pp. 445–455, 2012. Available: <u>https://doi.org/10.1080/19439342</u>. 2012.711342
- [68] H. Crompton and D. Burke, "The use of mobile learning in higher education: A systematic review", *Computers & Education*, vol. 123, pp. 53–64, 2018. Available: <u>https://doi.org/ 10.1016/j.compedu.2018.04.007</u>
- [69] G. Duman, G. Orhon and N. Gedik, "Research trends in mobile assisted language learning from 2000 to 2012", *ReCALL*, vol. 27, no. 2, pp. 197–216, 2014. Available: <u>https://doi.org/10.1017/ S0958344014000287</u>
- [70] N. Salkind, Encyclopedia of research design. London: SAGE, 2010. <u>https://doi.org/10.4135/9781412961288</u>
- [71] Y. R. Lee, M. Md. Yunus and A. Suliman, "The influence of social media on spelling skills among primary school students", *International Journal of Innovation, Creativity* and Change, vol. 7, no. 6, pp. 284–297, 2019. Available: <u>https://www.ijicc.net/images/ vol7iss6/7619_Rou_2019_E_R.pdf</u>
- [72] Y. L. P. K. Abdullah, A, N. Abdullah, P. Govindasamy, and M. Md. Yunus, "Learn idioms the fun and mobile way with Quizizz", *International Journal of Scientific & Technology Research*, vol. 8, no. 12, pp. 1749–1752, 2019. Available: <u>http://www.ijstr.org/final-print/</u> dec2019/Learn-Idioms-The-Fun-And-Mobile-Way-With-Quizizz.pdf

- [73] P. Govindasamy, M. Yunus and H. Hashim, "Mobile assisted vocabulary learning: Examining the effects on students' vocabulary enhancement", *Universal Journal of Educational Research*, vol. 7, no. 12, pp. 85–92, 2019. Available: <u>https://doi.org/10.13189/ujer.2019.071911</u>
- [74] S. Ng, M. Azlan, A. Kamal and A. Manion, "A quasi-experiment on using guided mobile learning interventions in ESL classrooms: Time use and academic performance", *Education* and Information Technologies, vol. 25, no. 6, pp. 4699–4719, 2020. Available: <u>https://doi.org/10.1007/s10639-020-10191-7</u>
- [75] T. Rajendran and M. Md. Yunus, "A systematic literature review on the use of mobile assisted language learning (MALL) for enhancing speaking skills among ESL and EFL learners", *International Journal of Academic Research in Progressive Education and Devel*opment, vol. 10, no. 1, pp. 586–609, 2021. Available: <u>https://doi.org/10.6007/IJARPED/ v10-i1/8939</u>
- [76] H. Hashim, M. Md. Yunus and M. A. Embi, "Learning through mobile: exploring the views of Polytechnic ESL learners", *TLEMC (Teaching and Learning English in Multicultural Contexts)*, vol. 2, no. 1, pp. 56–62, 2018. Available: <u>http://jurnal.unsil.ac.id/index.php/ tlemc/article/view/490</u>
- [77] T. W. Wen et al., "Effectiveness of mobile assisted language learning towards students' achievement and motivation in learning English prepositions", In 2019 IEEE International Conference on Engineering, Technology and Education (TALE), 2019, pp. 1–5. Available: <u>https://doi.org/10.1109/TALE48000.2019.9225865</u>
- [78] A. J. Ahmad Tajuddin, R. Romly, H. Ab Hamid, W. N. Wan Salam and N. I. Jamaludin@ Noordin, "Keeping up with digital trends: Reflecting students' experiences using MP3 experiment in non-formal listening activity", *International Journal of Innovative Technology and Exploring Engineering (IJITEE)*, vol. 8, no. 7S2, pp. 86–90, 2019. Available: <u>https://www. ijitee.org/wp-content/uploads/papers/v8i7s2/G10140587S219.pdf</u>
- [79] S. Azar and N. H. Iskandar Tan, "The application of ICT techs (mobile-assisted language learning, gamification, and virtual reality) in teaching English for secondary school students in Malaysia during Covid-19 pandemic", *Universal Journal of Educational Research*, vol. 8, no. 11C, pp. 55–63, 2020. Available: <u>https://www.hrpub.org/download/20201130/</u> UJERC7-19591547.pdf; https://doi.org/10.13189/ujer.2020.082307
- [80] F. D. Deris and N. S. A Shukor, "Vocabulary learning through mobile apps: A phenomenological inquiry of student acceptance and desired apps features", *International Journal of Interactive Mobile Technologies*, vol. 13, no. 7, pp. 129–140, 2019. Available: <u>https://doi.org/10.3991/ijim.v13i07.10845</u>
- [81] T. Teo, T. Doleck, P. Bazelais and D. J. Lemay, "Exploring the drivers of technology acceptance: A study of Nepali school students", *Educational Technology Research and Development*, vol. 67, no. 2, pp. 495–517, 2019. Available: <u>https://doi.org/10.1007/s11423-019-09654-7</u>
- [82] N. Abdallah, O. Abdallah and O. M Bohra, "Factors affecting mobile learning acceptance in higher education: An empirical study", *International Journal of Advanced Computer Science and Applications*, vol. 12, no. 4, pp. 664–671, 2021. Available: <u>https://thesai.org/ Downloads/Volume12No4/Paper_82-Factors_Affecting_Mobile_Learning_Acceptance.</u> pdf; https://doi.org/10.14569/IJACSA.2021.0120482
- [83] H. Hashim, M. Md. Yunus and N. Y. K. Zakaria, "Pre-university ESL learners' attitude towards m-learning: An investigation on the moderating role of gender", *Asian EFL Journal*, vol. 27, no. 4.3, pp. 40–62, 2020. Available: <u>https://www.asian-efl-journal.com/wp-content/uploads/AEJ-Volume-27-Issue-4.3-October-2020.pdf</u>

- [84] M. A. Almaiah and O. A. Alismaiel, "Examination of factors influencing the use of mobile learning system: An empirical study", *Education and Information Technologies*, vol. 24, no. 1, pp. 885–909, 2019. Available: <u>https://doi.org/10.1007/s10639-018-9810-7</u>
- [85] A. N. Zulfakar and N. A Zabidi, "Perceptions and attitudes of the use of mobile phones in English language learning", *Journal Wacana Sarjana*, vol. 4, no. 3, pp. 1–11, 2020. Available: <u>http://spaj.ukm.my/jws/index.php/jws/article/view/326</u>
- [86] L. Briz-Ponce, A. Pereira, L. Carvalho, J. A. Juanes-Méndez and F. J. García-Peñalvo, "Learning with mobile technologies–Students' behavior", *Computers in Human Behavior*, vol. 72, pp. 612–620, 2017. Available: <u>https://doi.org/10.1016/j.chb.2016.05.027</u>
- [87] A. Goswami and S. Dutta, "Gender differences in technology usage—A literature review", Open Journal of Business and Management, vol. 4, no. 1, pp. 51–59, 2015. Available: <u>https://doi.org/10.4236/ojbm.2016.41006</u>
- [88] N. Bawa and Z. Suleiman, "Gender dimension from effect of mobile instructional app on undergraduates' performance in Economics", *International Journal of Innovative Technol*ogy Integration in Education, vol. 4, no. 1, pp. 118–124, 2020. Available: <u>https://ijitie.aitie.org.ng/index.php/ijitie/article/view/153</u>
- [89] N. Annamalai and J. A Kumar, "Understanding smartphone use behavior among distance education students in completing their coursework in English: A mixed-method approach", *The Reference Librarian*, vol. 61, no. 3–4, pp. 199–215, 2020. Available: <u>https://doi.org/ 10.1080/02763877.2020.1815630</u>
- [90] B. Klimova, "Impact of mobile learning on students' achievement results", *Education Sciences*, vol. 9, no. 2, pp. 90, 2019. Available: <u>https://doi.org/10.3390/educsci9020090</u>
- [91] K. Nikolopoulou, "Secondary education teachers' perceptions of mobile phone and tablet use in classrooms: benefits, constraints and concerns", *Journal of Computers in Education*, vol. 7, no. 2, pp. 257–275, 2020. Available: <u>https://doi.org/10.1007/s40692-020-00156-7</u>
- [92] F. Nami, "Educational smartphone apps for language learning in higher education: Students' choices and perceptions", *Australasian Journal of Educational Technology*, vol. 36, no. 4, pp. 82–95, 2020. Available: <u>https://doi.org/10.14742/ajet.5350</u>
- [93] M. S. H. Khan, B. O Abdou, J. Kettunen and S. Gregory, "A phenomenographic research study of students' conceptions of mobile learning: An example from higher education", *Sage Open*, vol. 9, no. 3, pp. 1–17, 2019. Available: <u>https://doi.org/10.1177/2158244019861457</u>
- [94] R. Kumar, *Research Methodology: A Step-By-Step Guide for Beginners*. London: Sage Publications Ltd, 2018.
- [95] C. Yue and X. Xu, "Review of quantitative methods used in Chinese educational research, 1978–2018", ECNU Review of Education, vol. 2, no. 4, pp. 515–543, 2019. Available: https://doi.org/10.1177/2096531119886692
- [96] E. R. Babbie, The Practice of Social Research. London: Cengage Learning, 2020.
- [97] J. W. Creswell, *Educational Research: Planning, Conducting, And Evaluating Quantitative and Qualitative Research*, 4th ed. Boston, MA: Pearson, 2012.
- [98] M. Seyfried and F. Reith, "Mixed methods for research into higher education: solving the problem of institutionalised introspection?", *Theory and Method in Higher Education Research*, pp. 111–127, 2019. Available: <u>https://doi.org/10.1108/S2056-375220190000005008</u>
- [99] R. Timans, P. Wouters and J. Heilbron, "Mixed methods research: what it is and what it could be", *Theory and Society*, vol. 48, no. 2, pp. 193–216, 2019. Available: <u>https://doi.org/10.1007/s11186-019-09345-5</u>
- [100] J. W. Creswell and C. V. L. Plano, "Designing and Conducting Mixed Methods Research. Thousand Oaks, CA: Sage Publications Ltd, 2017.

- [101] J. F. Molina-Azorin, "Mixed methods research: An opportunity to improve our studies and our research skills", *European Journal of Management and Business Economics*, vol. 25, pp. 37–38, 2016. Available: <u>https://doi.org/10.1016/j.redeen.2016.05.001</u>
- [102] C. B. Gibson, "Elaboration, generalisation, triangulation, and interpretation", Organizational Research Methods, vol. 20, no. 2, pp. 193–223, 2016. Available: <u>https://doi.org/10.1177/1094428116639133</u>
- [103] C. Anderson, "Presenting and evaluating qualitative research", American Journal of Pharmaceutical Education, vol. 74, no. 8, pp. 1–7, 2010. Available: <u>https://doi.org/10.5688/ aj7408141</u>
- [104] H. Luo, "Qualitative research on educational technology: Philosophies, methods and challenges", *International Journal of Education*, vol. 3, no. 2, pp. 1–16, 2011. Available: <u>https://doi.org/10.5296/ije.v3i2.857</u>
- [105] A. Lenger, "The rejection of qualitative research methods in economics", *Journal of Economic Issues*, vol. 53, no. 4, pp. 946–965, 2019. Available: <u>https://doi.org/10.1080/002136</u> 24.2019.1657748
- [106] A. H. Hilal and S. S. Alabri, "Using NVivo for data analysis in qualitative research", *Inter-national Interdisciplinary Journal of Education*, vol. 2, no. 2, pp. 181–186, 2013. Available: http://www.iijoe.org/v2/IIJOE_06_02_02_2013.pdf; https://doi.org/10.12816/0002914
- [107] B. Johnson and L. Christensen, *Educational Research*, 4th ed. Los Angeles, CA: Sage Publications Ltd, 2012.
- [108] B. Miles and A. M. Huberman, *Qualitative Data Analysis*. Thousand Oaks, CA: Sage Publications Ltd, 2009.
- [109] C. Erlingsson and P. Brysiewicz, "A hands-on guide to doing content analysis", *African Journal of Emergency Medicine*, vol. 7, no. 3, pp. 93–99, 2017. Available: <u>https://doi.org/10.1016/j.afjem.2017.08.001</u>
- [110] S. A. Saleh and S. A. Bhat, "Mobile learning: A systematic review", *International Journal of Computer Applications*, vol. 114, no. 11, pp. 1–5, 2015. Available: <u>https://doi.org/10.5120/20019-1406</u>
- [111] V. Persson and J. Nouri, "A systematic review of second language learning with mobile technologies", *International Journal of Emerging Technologies in Learning*, vol. 13, no. 2, pp. 188–210, 2018. Available: <u>https://doi.org/10.3991/ijet.v13i02.8094</u>
- [112] M. Idris and A. Hussain, "Deploying mobile trend in higher education institutions in Malaysia", *International Journal of Engineering & Technology*, vol. 8, no. 1.12, pp. 59–62, 2019. Available: <u>https://www.sciencepubco.com/index.php/ijet/article/view/28847/15308</u>
- [113] R. Karim, A. Adnan, M. Salim, S. Kamarudin and A. Zaidi, "Education innovations through mobile learning technologies for the Industry 4.0 readiness of tertiary students in Malaysia. In (vol. 917, no. 1, p. 012022).", in *IOP Conference Series: Materials Science and Engineering*, 2020. Available: https://doi.org/10.1088/1757-899X/917/1/012022
- [114] J. Kankok, A. S. Ambotang and N. F. A. Kariming, "Mobile learning adoption: A perspective from a form six students in Sabah, Malaysia", *Malaysian Journal of Social Sciences and Humanities (MJSSH)*, vol. 5, no. 12, pp. 314–332, 2020. Available: <u>https://doi.org/10.47405/mjssh.v5i12.563</u>
- [115] R. Rahamat, "Designing mobile learning: Empirical journey to reality in Malaysian secondary school context", *EduLite: Journal of English Education, Literature and Culture*, vol. 4, no. 1, pp. 13–24, 2019. Available: <u>https://doi.org/10.30659/e.4.1.13-24</u>
- [116] H. Crompton, "A historical overview of mobile learning: Toward learner-centered education," in *Handbook of Mobile Learning*, Z. L. Berge and L. Y. Muilenburg, Eds. Florence: KY: Routledge, 2013, pp. 3–14.

- [117] Malaysian Ministry of Education, Malaysia Education Blueprint 2013–2025. Putrajaya: MOE, 2013.
- [118] A. Shahzad, R. Hassan, A. Y. Aremu, A. Hussain and R. N. Lodhi, "Effects of COVID-19 in E-learning on higher education institution students: the group comparison between male and female", *Quality & Quantity*, vol. 55, no. 3, pp. 805–826, 2021. Available: <u>https://doi.org/10.1007/s11135-020-01028-z</u>
- [119] Z. Ali Othman, M. M. S. Meor Sepli, U. A. Mokhtar and Y. H. Yeop, "BYOD implementation model in Malaysian schools: The perception and readiness of parents, schools, and teachers," *International Journal of Advanced and Applied Sciences*, vol. 7, no. 6, pp. 57–68, 2020. Available: <u>https://doi.org/10.21833/ijaas.2020.06.008</u>
- [120] N. F. Sa'don, H. M. Dahlan, A. Ibrahim and N. Fadzleen, "Usage of mobile learning in Malaysian secondary education: Stakeholders' view", *Journal of Information Systems Research and Innovation*, vol. 6, pp. 42–50, 2014. Available: <u>https://seminar.utmspace.edu.my/jisri/download/f_vol6feb2014_fullpaper/pub6_mobilelearninginmalaysiansecondaryeducation.pdf</u>
- [121] C. M. Chao, "Factors determining the behavioral intention to use mobile learning: An application and extension of the UTAUT model", *Frontiers in Psychology*, vol. 10, no. 1652, pp. 1–14, 2019. Available: <u>https://doi.org/10.3389/fpsyg.2019.01652</u>
- [122] F. Weng, R. J. Yang, H. J. Ho and H. M. Su, "A TAM-based study of the attitude towards use intention of multimedia among school teachers", *Applied System Innovation*, vol. 1, no. 3, pp. 36, 2018. Available: <u>https://doi.org/10.3390/asi1030036</u>
- [123] R. A. Abdulrahman and E. Benkhelifa, "A systematic literature review on mobile learning in Saudi Arabia," In 2017 IEEE/ACS 14th International Conference on Computer Systems and Applications (AICCSA), 2017, pp. 1354–1361. <u>https://doi.org/10.1109/ AICCSA.2017.211</u>
- [124] H. Crompton, D. Burke, K. H. Gregory and C. Gräbe, "The use of mobile learning in science: A systematic review", *Journal of Science Education and Technology*, vol. 25, no. 2, pp. 149–160, 2016. Available: <u>https://doi.org/10.1007/s10956-015-9597-x</u>
- [125] S. B. Bhaskar, "Concealing research outcomes: Missing data, negative results and missed publications", *Indian Journal of Anesthesia*, vol. 61, no. 6, pp. 435–453, 2017. Available: <u>https://doi.org/10.4103/ija.IJA_361_17</u>
- [126] L. Echevarría, A. Malerba and V. Arechavala-Gomeza, "Researcher's perceptions on publishing "negative" results and open access", *Nucleic Acid Therapeutics*, vol. 31, pp. 185– 189, 2020. Available: <u>https://doi.org/10.1089/nat.2020.0865</u>
- [127] E. Granqvist, "Why science needs to publish negative results (Authors' Update)", *Elsevier*: com, 2015. [Online]. Available: <u>https://www.elsevier.com/connect/authors-update/why-science-needs-to-publish-negative-results</u>. [Accessed: 14- May- 2021].
- [128] L. van Hilten, "Why it's time to publish research "failures", *Elsevier Connect*, 2015. [Online]. Available: <u>https://www.elsevier.com/connect/scientists-we-want-your-negative-results-too</u>. [Accessed: 13- May- 2021].
- [129] S. B. Merriam and E. J. Tisdell, *Qualitative Research: A Guide to Design and Implemen*tation. San Francisco, CA: Jossey-Bass, 2016.
- [130] A. Osanloo and C. Grant, "Understanding, selecting, and integrating a theoretical framework in dissertation research: Creating the blueprint for your "house"", *Administrative Issues Journal: Connecting Education, Practice, and Research*, vol. 4, no. 2, pp. 7, 2016. Available: <u>https://doi.org/10.5929/2014.4.2.9</u>
- [131] C. Kivunja, "Distinguishing between theory, theoretical framework, and conceptual framework: A systematic review of lessons from the field", *International Journal of Higher Education*, vol. 7, no. 6, pp. 44–53, 2018. Available: <u>https://doi.org/10.5430/ijhe.v7n6p44</u>

8 Authors

Nursufyana Ulfa Che Mustaffa has a Bachelor of Education in Teaching English as a Foreign language (TEFL) from the Queensland University of Technology, Australia and a Master of Education specialisation in English Language Teaching from University Utara Malaysia (UUM). She is presently enrolled in UUM, pursuing a PhD in English Language Teaching. Her research interests include English Language Teaching (ELT) and mobile learning, with a specific emphasis on the use of mobile technologies to facilitate English language teaching and learning among Malaysian primary school pupils. She may be contacted by e-mail at <u>n.sufyanaulfa@gmail.com</u>.

Siti Nazuar Sailin is a Senior Lecturer of Educational Technology in the School of Education, College of Arts and Sciences, UUM. Presently, she is the Deputy Dean of Awang Had Salleh Graduate School (AHSGS) in UUM. She graduated from UUM with a Bachelor Degree with Honours in Information Technology (IT) and a Master of Science in Information Technology (IT). She also holds a Master of Information and Communication Technology in Education and a PhD in Educational Technology from Monash University, Australia. She is also a member of the editorial board of Practitioner Research, a peer-reviewed journal published by UUM Press. Her expertise and interests include teachers' professional development, educational technology, and online and blended learning. She may be reached by e-mail at sitinaz@uum.edu.my.

Article submitted 2021-11-07. Resubmitted 2021-12-25. Final acceptance 2021-12-25. Final version published as submitted by the authors.